
\documentclass[11pt]{article}
% \usepackage[lucidasmallscale,nofontinfo]{lucimatx}
\usepackage{pifont}
\usepackage{booktabs}
\usepackage[margin=1.0in]{geometry}

\usepackage{xcolor}
\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}
\usepackage[colorlinks=true,linkcolor=webgreen,filecolor=webbrown,citecolor=webgreen]{hyperref}


\usepackage{pgf,tikz,xcolor} % for tikz graphs and colors
\usetikzlibrary{decorations.text} % decorations.text for the text along path feature

\def\BibTeX{{\rm B\kern-.05em{\sc i\kern-.025em b}\kern-.08em T\kern-.1667em\lower.7ex\hbox{E}\kern-.125emX}}

% \usepackage[annotation=true,sorting=ydnt,backend=biber]{biblatex}
\usepackage[sorting=none,backend=biber]{biblatex}
\addbibresource{nlp.bib}


\renewbibmacro*{finentry}{%
  \setunit{\addperiod\addnbspace\textendash\space}%
  \printfield{annotation}%
  \finentry}


\defbibheading{bibliography}{%
  \subsubsection*{Annotated Bibliography}}

\renewenvironment*{thebibliography}
  {\list{}{%
     \setlength{\leftmargin}{1.5em}%
     \setlength{\itemindent}{-\leftmargin}%
     \setlength{\itemsep}{0.5\baselineskip}%
     \setlength{\parsep}{0pt}}}
  {\endlist}
%

\title{Your Topic Goes Here: An Annotated Bibliography \\ East Carolina University}
\author{Your Name Goes Here}
\date{ }

\begin{document}
\maketitle
\tableofcontents
\thispagestyle{empty}

\newpage

\section{Introduction}

This bibliography is intended for beginning students of Natural Language  Processing (NLP). Hence, its focus is intentionally broad.


\section{Facets of NLP}

There are several facets to NLP including those that unify various aspects (general), language modeling, tagging, parsing, information extraction, machine translation, language generation, dialog systems, question-answering systems, discourse understanding, clustering, and summarization. 



\subsection{General}

These are general works that are of interest to beginning students. More to come.

\refsection
\nocite{Abend2011,Belogay2012,Bird2009,GATE,Goldwater2011,Goyal2010,Indurkhya2010,Ingersoll2012,Madnani2007,Manning1999,Norvig2009,Perkins2010,Popel2010,Stanford-NLP,Tetreault2008,openNLP}


\printbibliography
\endrefsection



\subsection{NLP Education}

These are general works that are of interest to NLP learning and teaching.

\refsection
\nocite{Bird2008,Burstein2005,Church2012,Cushing2009,Freedman2005,Freedman2008,Jurafsky2009,Kano2009,Klein2005a,Lee2002,Liddy2005,Madnani2008,Tetreault2009,Tetreault2010,Tetreault2011,Xia2008}
\printbibliography
\endrefsection


\subsection{Language Modeling}

These works discuss approaches to modeling and evaluating natural languages. 

\refsection
\nocite{Belz2011,Brants2012,Dimitriadis2010,Goyal2009,Mokhov2010,Rago2009,Weissenbacher2006}
\printbibliography
\endrefsection


\subsection{Tagging and Annotation}

These works discuss approaches to tagging natural language text. 

\refsection
\nocite{Ekbal2011,Hovy2010,Stenetorp2012,Zhang2010a}
\printbibliography
\endrefsection



\subsection{Parsing}

These works discuss approaches and algorithms for parsing natural language text.

\refsection
\nocite{Abdalgader2012,Davidov2009,Klein2005,Navigli2009,Subramaniam2009,Wang2011e}
\printbibliography
\endrefsection



\subsection{Information Extraction}

These works discuss approaches and algorithms for extracting information from natural language. 

\refsection
\nocite{Assal2010,Clark2008,Getoor2012,Guo2009,Jain2009a,Watrin2011,Wimalasuriya2010}
\printbibliography
\endrefsection




\subsection{Machine Translation}

These works discuss approaches and algorithms for machine translation of natural languages. 


\refsection
\nocite{Callison-Burch2009,Heger2010,Zbib2010}
\printbibliography
\endrefsection




\subsection{Language Generation}

These works discuss approaches and algorithms for generating natural language.

\refsection
\nocite{Dannells2010,Rieser2010}
\printbibliography
\endrefsection



\subsection{Dialog Systems}

These works discuss approaches and algorithms for developing dialog systems. More to come.

\refsection
\nocite{Griol2014}
\printbibliography
\endrefsection



\subsection{Question-answering Systems}

These works discuss approaches and algorithms for question-answering systems. 

\refsection
\nocite{Anderson2012,Andrenucci2005,Shtok2012}
\printbibliography
\endrefsection



\subsection{Question-generation Systems}

These works discuss approaches and algorithms for automatic generation of test questions for learning and assessment. 


\refsection
\nocite{Agarwal2011,Boyer2010,Heilman2010,Therrien2008}
\printbibliography
\endrefsection



\subsection{Discourse Understanding}

These works discuss approaches and algorithms for developing discourse understanding systems. 


\refsection
\nocite{Hernault2010}
\printbibliography
\endrefsection



\subsection{Clustering}

These works discuss approaches and algorithms for clustering natural language text. 

\refsection
\nocite{Navigli2011}
\printbibliography
\endrefsection




\subsection{Summarization}

These works discuss approaches and algorithms for summarizing single and multiple documents. 


\refsection
\nocite{Louis2010,OTS}
\printbibliography
\endrefsection




\subsection{NLP Applications}

These works discuss practical applications of NLP. 


\refsection
\nocite{Kant2012,Lau2012,Laws2011,Leroy2012,LingPipe,McKenzie2010,Mihalcea2011,Paltoglou2012,Russell2011,Sabou2012,Sharma2010}
\printbibliography
\endrefsection



\section{Peer Feedback and Revision} \label{sec:peer}

Remember that the goal of peer feedback is not only to point out shortcomings in the paper, but more importantly to suggest ways to improving the quality of the paper (writing style, flow, grammar, technical accuracy). 

\noindent Name of the person who provided feedback: Peer's name goes here.

Following are the specific suggestions made by the peer:

\begin{enumerate}

   \item suggestion 1 description

   \item suggestion 2 description

   \item suggestion $n$ description

\end{enumerate}

In the following, discuss how the peer's suggestions for improvement are incorporated into the document.

\begin{enumerate}

   \item Explain how peer's suggestion 1 is addressed

   \item Explain how peer's suggestion 2 is addressed

   \item Explain how peer's suggestion $n$ is addressed

\end{enumerate}


\section{Instructor Feedback and Revision} \label{sec:instructor}

This section should list feedback provided by the instructor and discuss how you have incorporated the feedback in improving the paper.


Following are the specific suggestions made by the instructor:

\begin{enumerate}

   \item suggestion 1 description

   \item suggestion 2 description

   \item suggestion $n$ description

\end{enumerate}

In the following, discuss how the instructor's suggestions for improvement are incorporated into the document.

\begin{enumerate}

   \item Explain how instructor's suggestion 1 is addressed

   \item Explain how instructor's suggestion 2 is addressed

   \item Explain how instructor's suggestion $n$ is addressed

\end{enumerate}


\section{Self-assessment} \label{sec:assessment}

This assignment will receive zero points if the required \LaTeX{}\ template is not used. 

\begin{center}
\begin{tabular}{p{3in}rr}
\toprule

\textbf{Rubric line item} & \multicolumn{1}{c}{\textbf{Max possible points}} & \multicolumn{1}{c}{\textbf{Earned points}} \\ 
\midrule

Read the paper titled ``Scaffolding Beginning Research Students Using Open Source Tools'' and used its ideas in completing this assignment. & 10 & nn \\ \midrule

Consulted ACM Digital Library, IEEE Computer Society Digital Library, SiteSeerX, {\sc Mendeley}, Google Scholar, Ultimate Research Assistant, and other online sources for identifying and collecting bibliography sources.  & 10 & nn \\ \midrule

Several facets of the assigned topic has been identified and the annotated bibliography is organized according to these facets. & 20 & nn \\ \midrule

For each bibliography source, included are: summary in the form of abstract, assessment, and reflection.  & 30 & nn \\ \midrule

\LaTeX{}\ and \BibTeX{}\ files follow standard conventions and are easy to read and maintain.  & 5 & nn \\ \midrule

Peer feedback is solicited and incorporated. & 5 & nn \\ \midrule

Instructor feedback is solicited and incorporated. & 5 & nn \\ \midrule

Writing is of professional quality and is free from grammatical and syntactic errors  & 10 & nn \\ \midrule

Self-assessment is performed. & 5 & nn \\ \midrule

& & \\

\textbf{Total points} & 100 & nn \\
\bottomrule
\end{tabular}
\end{center}


\end{document}